Page 1 of 6

CS-251A Data Structur es and Algorithms

Lectures, Homework and Play Week I. Sept 2

A. Lectur e #1:

1. Introduce Course 2. Define computer science, algorithm, program, machine 3. Discuss analog versus digital computers 4. Gravity computer (centroid of a polygon) 5. Abacus 6. Knotted string computer 7. Prove Pythagoras’ theorem 8. Prove the converse of Pythagoras’ Theorem 9. Straight edge and compass computer (Euclidean operations) 10. Collapsing compass and Euclid’s second proposition 11. Squaring a rectangle

B. Problem Assignment #1

1. Squaring a convex pentagon with straight edge and compass 2. Transferring and orthogonalizing a segment with the ollapsing compass.

C. Reading Assignment

1. Text: Preface and Chapter #1 2. Handout: A New Look at Euclid’s Second Proposition

D. Suggested Play

1. Applet of Francois Labelle

Week II. Sept 7 and 9

A. Lectur e #1: 1. B. Lectur e #2: 1.